Author's response to reviews

Title: The effect of spinal curvature on the photogrammetric assessment on static balance in elderly women

Authors:

Justyna Drzal-Grabiec (justyna.drzal.grabiec@wp.pl)
Maciej Rachwal (rachwal@univ.rzeszow.pl)
Justyna Podgórńska-Bednarz (j.e.podgorska@gmail.com)
Justyna Rykala (justynarykala@tlen.pl)
Slawomir Snela (ssnela@szpital2.rzeszow.pl)
Aleksandra Truszczynska (aleksandra.rapala@wp.pl)
Zbigniew Trzaskoma (zbyszek.trzaskoma@awf.edu.pl)

Version: 5 Date: 16 May 2014

Author's response to reviews:

Dear Editorial Team,
Thank you very much for acceptance of the paper.
I have confirmed that the patient had given their consent for their image to be published. I have include a statement in the manuscript to confirm this at the end of the paper, and the change is marked on red.

Kind regards
Aleksandra Truszczynska